Author’s response to reviews

Title: Patients' and caregivers' experiences of using continuous glucose monitoring to support diabetes self-management: qualitative study

Authors:

Julia Lawton (j.lawton@ed.ac.uk)
Maxine Blackburn (m.blackburn@ed.ac.uk)
Janet Allen (ja385@medschl.cam.ac.uk)
Fiona Campbell (Fiona.campbell26@nhs.net)
Daniela Elleri (Daniela.Elleri@nhslothian.scot.nhs.uk)
Lalantha Leelarathna (Lalantha.Leelarathna@cmft.nhs.uk)
David Rankin (a.d.rankin@ed.ac.uk)
Martin Tauschmann (mt614@medschl.cam.ac.uk)
Hood Thabit (Hood.Thabit@cmft.nhs.uk)
Roman Hovorka (rh347@cam.ac.uk)

Version: 2 Date: 05 Feb 2018

Author’s response to reviews:

Many thanks indeed for provisionally accepting our manuscript. Below we have outlined how we have addressed the final set of essential revisions.

1. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Response: we have now provided a list of all abbreviations used in our manuscript before the Declarations section. We have also defined all abbreviations at first use in the text.

2. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state n/a. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing...
it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

Response: we now provide a rationale for why it is inappropriate to share our raw data (interview transcripts) in the Availability of data and materials section, and stated that raw data are available from the corresponding author on reasonable request.

3. Please note that consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity.

Response: noted, we have changed the text.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: we have uploaded a clean version.